Author's response to reviews

Title: Prospective study of asbestos-related diseases incidence cases in primary health care in an area of Barcelona province

Authors:

Magdalena Rosell Murphy (mrosell@idiapjgol.org)
Rafael Abós- Herràndiz (24025rah@comb.cat)
Josep Tarrés (7893ito@comb.cat)
Xavier Martínez-Artés (xmartinezar@yaho.es)
Isabel García-Allas (26584iga@comb.cat)
Ilona Krier (ilonakrier@yaho.es)
Gloria Cantarell (gcantarell@tauli.cat)
Miguel Gallego (mgallego@tauli.cat)
Ramon Orriols (rorriols@vhebron.net)
Constança Albertí (Constanca.alberti@gencat.cat)

Version: 3 Date: 25 March 2010

Author's response to reviews:

1. I have done the following changes from the previous version:
2. I have arranged the order of references that was not correct; in particular number 11 and 19.
3. have removed the manuscript number form the title page
4. Rename d the Methos/design in abstract and manuscript
5. Removed the highlighting in the text
6. Placed the appendix (1, 2 and 3) before the acknowledgements section of the manuscript
7. Check the typography

I am attaching the new version with all the changes stated